Author's response to reviews

Title: The State of the Research for Health Environment in the Ministries of Health of the Economic Community of the West African States (ECOWAS)

Authors:

Issiaka Sombie (isombie@wahooas.org)
Jude Aidam (jaidam@wahooas.org)
Blahima Konaté (kobla70@hotmail.com)
Télesphore D Somé (telesome@hotmail.com)
Stanislas S Kambou (skambou@wahooas.org)

Version: 3 Date: 30 August 2013

Author's response to reviews:

Thanks again to Dr. RM for her kind and detailed comments.

We have further taken the opportunity to improve the flow and clarified some seeming ambiguities. We have also worked to improve the grammar and the writing.

Introduction: We have provided the 2 frameworks referred to that seems to have left the text hanging as diagrams

Methods: this is certainly not a case study and therefore the correction has been made.

Results: The grammar has been improved through out the whole article.

Thank you once again for your diligent and detailed review. This has been really helpful.